Author’s response to reviews

Title: Laparoscopic colonic resection for splenic flexure cancer: our experience.

Authors:

Andrea Pisani Ceretti (andreapisaniceretti@yahoo.it)
Nirvana Maroni (nirvana.maroni@yahoo.it)
Matteo Sacchi (matteo.sacchi@humanitas.it)
Stefano Bona (stefano.bona@humanitas.it)
Maria Rachele Angiolini (rakele86@gmail.com)
Paolo Pietro Bianchi (paolopietro.bianchi@fastwebnet.it)
Enrico Opocher (enrico.opocher@unimi.it)
Marco Montorsi (marco.montorsi@unimi.it)

Version: 11 Date: 10 June 2015

Author’s response to reviews: see over
Dear Magdalena Morawska,

We would like to thank You again and all the Reviewers for the in-depth revision of our manuscript and for your patience. We are now resubmitting the manuscript, which has been further revised according to the Reviewers’ comments; you can find highlighted in the manuscript the main changes done. Please see below our point-to-point reply to the Reviewers comments, including the most recent.

We are grateful to the Reviewers who provided us with helpful insights to improve our paper.

We hope that You will find the manuscript suitable for publication in your journal now.

Yours sincerely,

Andrea Pisani Ceretti, M.D.

Editorial Requirement

Further issues reported in your most recent communication (2015/06/08)

(1) Competing Interests: Please change the title of the "Author Disclosures" section to "Competing Interests"
   We did

(2) Authors’ Contributions: Please use only initials of authors (for example APC instead of Andrea Pisani Ceretti) instead of full names.
   We did